Massachusetts Institute of Technology Handout 11 6.915: Network and Computer Security October 3, 1995 Professor Ronald L. Rivest

Readings for Lecture 8

Attached is the following list of papers

ffl A Logic of Authentication by Michael Burrows, Martin Abadi and Roger Needham [1].

ffl EC plans encryption rules from Nature [2]

References

[1] Michael Burrows, Martin Abadi and Roger Needham, A Logic of Authentication, ACM Transactions on Computer Systems, Volume 8, No. 1, pp. 18{36, February 1990.

[2] J. Thorel, EC plans encryption rules in bid to police information superhighway, Nature, Vol.377, p.275, September 1995.